Author's response to reviews

Title: Effects of a physical education intervention on cognitive function in young children: randomized controlled pilot study.

Authors:

Abigail Fisher (a.fisher@ucl.ac.uk)
James ME Boyle (J.Boyle@strath.ac.uk)
James Y Paton (James.Paton@glasgow.ac.uk)
Phillip D Tomporowski (ptomporow@uga.edu)
Christine Watson (C.watson@glasgow.gov.uk)
John H McColl (John.McColl@glasgow.ac.uk)
John J Reilly (John.Reilly@glasgow.ac.uk)

Version: 6 Date: 24 October 2011

Author's response to reviews:

Apologies that it has taken me so long to complete the revision and resubmission of this manuscript. It would seem that a number of changes I had made to the online submission (author names, trial registration number) had not been saved in the submission system.